Author's response to reviews

Title: Pneumococcal pneumonia: differences according to blood culture results

Authors:

ALBERTO CAPELASTEGUI  
(ALBERTO.CAPELASTEGUISAIZ@OSAKIDETZA.NET)  
RAFAEL ZALACAIN (RAFAEL.ZALACAINJORGE@OSAKIDETZA.NET)  
AMAIA BILBAO (AMAIA.BILBAOGONZALEZ@OSAKIDETZA.NET)  
MIKEL EGURROLA (MIKEL.EGURROLAIZQUIERDO@OSAKIDETZA.NET)  
LUIS ALBERTO RUIZ ITURRIAGA  
(LUISALBERTO.RUIZITURRIAGA@OSAKIDETZA.NET)  
JOSE MARIA QUINTANA  
(JOSEMARIA.QUINTANALOPEZ@OSAKIDETZA.NET)  
AINHOA GOMEZ (AINHOA.GOMEZ@OSAKIDETZA.NET)  
CRISTOBAL ESTEBAN  
(CRISTOBAL.ESTEBANGONZALEZ@OSAKIDETZA.NET)  
PEDRO ESPAÑA  
(PEDROPABLO.ESPANAYANDIOLA@OSAKIDETZA.NET)

Version: 5  
Date: 15 July 2014

Author's response to reviews:

Responses to the specific comments of the reviewers.

1- Ethics committe name the name of the committees are included in the Methods section.

2- Consent all information about the consent is included in methods section.

3- Delete Financial/nonfinancial disclosures, role of sponsor and others contributions.

All of this sections have been deleted and now are included in Acknowledgements and competing interest.

Best regards

Dr. Alberto Capelastegui